Author’s response to reviews

Title: Hypertension in the Parsi community of Bombay: A study on prevalence, awareness and compliance to treatment

Authors:

Nadir E. Bharucha (nbharucha@vsnl.com)
Thomas Kuruvilla (olasa@vsnl.com)

Version: 2 Date: 27 Oct 2002

The first sentence under methods should read as 'We used a 1 in 4 random selection of subjects who were more or equal to 20 years of age' instead of 'We used a 1 in 4 random selection of subjects who were more than 20 years of age'